ITEMS REFERRED BY SENATE

Conditions of Appointment: Physicians Appointed in Clinical Departments

and Clinical Divisions of Basic Science Departments

Department of Biology: Amalgamation of Plant Sciences and Zoology

2002-03 Operating and Capital Budgets

First-Entry Undergraduate Enrolment for 2002-2003

First-Entry Undergraduate Enrolment for 2002-2003 - Affiliated Colleges

Report of the Provost's Undergraduate Program Review Committee

Report of the Academic Colleague on the 259th meeting of COU

Report of the Honorary Degrees Committee

Convocation Dates 2004

2001-2002 Winners: The Edward G. Pleva Award for Excellence in Teaching

2001-2002 Winner: The UWO Award for Excellence in Teaching by Part-Time Faculty

2001-2002 Winner: The Marilyn Robinson Award for Excellence in Teaching

2002 Winners: Hellmuth Prize for Achievement in Research

Announcements

FOR APPROVAL

1. <u>Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of</u> Basic Science Departments

Recommended:

That the Board of Governors approve revisions to Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments (1999) as detailed in Annex 1.

Highlights of Proposed Changes (in order of first appearance):

Table of Contents:

- 5.3 Change in name of Faculty Committee from Faculty Promotion and Tenure Committee to Faculty
 Appointments and Promotions Committee (this Committee will no longer deal with tenure cases which are now
 dealt with under the Collective Agreement)
- Elimination of Senate Committees on Promotion and Tenure and Senate Committee on Appeals (these Committees do not function under the Collective Agreement)
- 10.2 Faculty Grievance Committee (Clinical) changed to Faculty Grievance Committee (there is no longer a Faculty Grievance Committee under the Collective Agreement so the designation of "Clinical" is no longer required)
- 10.3 Senate Grievance Committee does not exist under the Collective Agreement so it is proposed to change this Committee to a University Grievance Committee
- Appendix I Limited Duties procedures are quite different under the Collective Agreement so it is proposed to amend Appendix I to be specific to physicians in the Faculty of Medicine & Dentistry
- Page 2: 2.2.5 and 2.2.6 Definitions for "Limited Term Appointment" and "Continuing Appointment" have been amended to include reference to retirement

Board of Governors APPENDIX VIII May 9, 2002 Page 2

- Page 3: 2.2.13 Honorary appointments new
 - 2.2.14 Joint Appointments joint appointments between two clinical departments will continue to fall under Policy 4.7, Manual of Administrative Policies and Procedures. Joint appointments between a basic and clinical department will be addressed by a Joint Committee Letter of Understanding or a subsequent Collective Agreement which may incorporate the Faculty's proposal for dealing with basic scientists in clinical departments.
- Page 5: 3.3.2.1 It is proposed that the file for an initial appointment with Continuing Appointment status would no longer require external referees or Senate Committee review (this is consistent with the requirements of Collective Agreement for an initial appointment with tenure)
- Page 6: Since 3.3.2.1 would no longer require external referees, 3.4.1 reinstates the requirement for external reviews when changing streams.
- Page 11: 4.3.10 Revised to eliminate Senate Committee review. Since the Provost's decision would now be substituted for the Senate Committee decision for Senate stream faculty, it is proposed that any appeal of the Provost's decision for Senate stream cases take the form of a grievance to the University Grievance Committee.
- Page 14: 5.3.1 Faculty Committee structure revised since former structure had to comply also with the former Conditions of Appointment structure
 - 5.3.1-5.3.6 It is proposed that the Dean may now delegate Chair duties to an Associate Dean.
 - 5.3.7.1 Reference to "Senate" has been deleted from Selection Committee.
- Page 15: 6.1.7 Specific notation has been added regarding academic colleague (consistent with past practice and requirements of Collective Agreement) See also 6.2.6 and 6.3.2.2.
 - 6.1.9 Senate Committee on Appeals would be replaced by University Grievance Committee.
- Page 22: 10.1.1 Addition of "Provost stream" since new 10.1.3 deals with new Senate Stream grievance process
- Page 23: 10.3.4 Grievances would now be submitted to the Director of Faculty Relations, rather than Secretary of Senate.
 - 10.3.6 Structure of the University Grievance Committee revised (former Senate Grievance Committee included the Provost as a member no longer appropriate since the Provost's decision may be grieved to this committee.)
- Appendix I: Changes to include Honorary Professor and the Appointments Committee may now delegate Limited Duties appointments to the Chair on an annual basis (consistent with Collective Agreement)

2. Department of Biology: Amalgamation of Plant Sciences and Zoology

Recommended: That the Board of Governors approve the amalgamation of the Departments of

Plant Sciences and Zoology in the Faculty of Science into a single department to

be called the Department of Biology, effective July 1, 2002.

Background:

Beginning in early 2001, the Departments of Plant Sciences and Zoology began discussing the possible merger of the departments into one. The two Chairs worked closely with each other and mutually consulted with each other through written communications and open meetings. The two departments have much in common, including the background and training of a majority of their faculty members, their research interests, pedagogical outlook and general philosophy.

Each department conducted a formal vote on the proposal to merge the two departments and in each case, there was greater than 85% support for the merger. The Council of the Faculty approved the merger in March. Subject to final approval by the Senate and the Board of Governors, the new department will be established on July 1, 2002. At that time all current members of UWOFA, UWOSA and PMA will be transferred to the new department, with no loss in seniority or job classification. All current academic programs, at both the undergraduate and graduate levels, will continue unchanged until such time as

Board of Governors APPENDIX VIII May 9, 2002 Page 3

formal changes are put forward following normal university procedures.

FOR INFORMATION

1. 2002-03 University Operating and Capital Budgets

The Operating and Capital Budgets and setting of tuition fees are in the purview of the Board of Governors. The Senate may provide its advice to the Board under the authority of Section 30.(f) of the *UWO Act*:

The Senate may ...

pass resolutions and make recommendations to the Board with respect to any matter connected with the administration of the University and the promotion of its affairs but this clause shall not be construed to subtract from the powers and duties conferred on the Board elsewhere in [the] Act.

At a meeting held on April 19, 2002, the Senate reviewed the proposed Operating and Capital Budgets and Tuition Fees. During discussion, three amendments relating to graduate tuition fees were proposed and subsequently defeated:

- 1. That Senate recommend to the Board that the proposed tuition fees for graduate students be rolled back to 2001-02 levels.
- 2. That Senate recommend to the Board that the proposed increases in graduate tuition fees be grandfathered in the case of students currently enrolled in graduate programs.
- 3. That Senate recommend to the Board that graduate international tuition fees be increased by the same rate as tuition paid by Canadian graduate students.

Following an extensive debate, Senate approved a resolution to provide advice to the Board of Governors, through the Vice-Chancellor, recommending the approval of the 2002-03 University Operating and Capital Budgets as presented.

2a. First-Entry Undergraduate Enrolment for 2002-2003

Senate, at its meeting of March 15, 2002, approved the following recommendations regarding first-entry undergraduate enrolment for 2002-2003:

- 1a) That the first-entry undergraduate enrolment objective for 2002-2003 be set at 4,500 students and that the approximate program-specific objectives be as described in **Annex 2**.
- b) That the 4,500 first-year student target be developed with an objective that 80% of the class be 'first-time' OAC students.
- c) That the Provost be granted permission to alter this enrolment objective, following consultation and approval from SUEPP, as might be appropriate given the particulars of a government funding announcement.
- d) That the entrance requirements be set by the Provost to achieve these objectives as a function of final admission numbers, qualifications of applicants, and estimates of the rate of offer confirmations in each program.

Board of Governors

May 9, 2002

APPENDIX VIII
Page 4

e) That no program-specific entrance requirement be set below 78% except where performance is a major element of the selection process. Confirmed admission to any program will be contingent upon a final average of no less than 74%.

2) That enrolment projections (Attachment 1-6) be used for purposes of tuition revenue estimates in the 2002-2003 University Budget.

Implementation:

Given the experience of recent years and current application figures, there is a possibility that minimum entrance requirements will be higher than 78% in 2002-2003.

The timing of the new application and offer process for OAC students (implemented in 1999-00) is such that early offers are made in March / April when less information is available regarding applicants than was the case prior to 1999-00.

The same framework will be applied to making early offers to OAC students in March / April, 2002, as was the case in the last year. Grade values (highlighted below in bold and italics) will be set as a function of final information regarding applicant qualifications in each program.

Offer will be made if:

- at least an 81% average on 3 OAC's and a blended Grade 11/12 average of at least 76%.
- at least an 81% average in Grade 11 and at least an 81% average in Grade 12, where fewer than 3 OAC's are available.
- at least an 86% average on 3 OAC's, where no blended Grade 11/12 average is available.
- at least an 81% average, where 6 OAC's are available.

The details of this matrix may vary slightly from program to program (e.g., in Music where auditions are an essential part of the admissions process).

Every effort will be made to maintain a single minimum standard of admission across all programs. However, a higher standard may be necessary in Nursing, BACS and Media, Information and Technoculture to meet enrolment targets. Our goal will be to set the same standard for Engineering as for Science, but given the special funding opportunities of the ATOP program, consideration will be given to setting the minimum entrance requirement in Engineering at 78% even if other programs have a marginally higher entrance requirement in 2002-2003.

Our objective is that a minimum of 80% of the first-year class be new OAC students. However, since the ultimate decision to register at Western resides with the student, the 80% figure must be viewed as an objective rather than an imperative.

As indicated in the Senate resolution of Spring of past years, consideration may be given in the admissions process to factors such as performance in program-relevant courses (e.g., mathematics and sciences in Science and Engineering), relevant extra curricular activities, and the academic record of the secondary school.

As was the case in 1999, 2000 and 2001, the period between the availability of initial application information (still not including final year, winter semester applicant grades) and the offer date does not permit an admissions proposal that includes detailed program-specific admissions criteria to be brought to Senate for consideration before initial offers were made.

An additional round of offers will be made in May 2002. Criteria used for those offers will be set as a function of additional information on applicant qualifications and program-specific patterns of acceptance of early offers, but will observe the general entrance requirements established in this recommendation.

Board of Governors APPENDIX VIII May 9, 2002 Page 5

2b. First-Entry Undergraduate Enrolment at the Affiliated Colleges, 2002-03

Senate, at its meeting of March 15, 2002, approved the following entrance requirements and processes for the Affiliated Colleges for 2002-2003:

Brescia University College: Brescia University College is targeting a first year class of 275 students in 2002-2003. As is the case at the Constituent University, Brescia University College will not set general entrance requirements until more complete information on applicant qualifications is available. On the basis of current information, however, the College tentatively is considering an entrance requirement of 74% for early offers in March- April 2002, and 72% for offers made in May 2002. Irrespective of final applicant qualification information, however, Brescia will set no program entrance requirement lower than 70%. The College will consider individual students with lower grades on the basis of supplementary information forms. No student will be admitted with a final OAC average of less than 68%.

Huron University College: Huron is targeting a first year class of 340 students in 2002-2003. As is the case at the Constituent University, Huron University College will not set general entrance requirements until more complete information on applicant qualifications is available. On the basis of currently available information, however, Huron is anticipating a minimum entrance requirement of 78% for early offers in March / April 2002. No student will be admitted with a final OAC average of less than 75%.

King's College. King's College is targeting a first year class of 895 students in 2002-2003. Timing of offers will follow the process set out by the Constituent University. On the basis of currently available information, and depending on the quality of the applicant pool, King's anticipates a minimum entrance average of 75% for Arts and Social Science and 80% for Childhood and Family Relations. No student will be admitted with a final OAC average of less than 70%.

All Affiliated Colleges will be bound to the entrance requirements established by the Constituent University for limited enrolment programs, including BACS, Media, Information and Technoculture, BHSc, and Kinesiology.

- 3. Report of the Provost's Undergraduate Program Review Committee See Annex 3.
- 4. Report of the Academic Colleague on the 259th meeting of the Council of Ontario Universities
 See Annex 4.
- 5. Report of the Honorary Degrees Committee
 See Annex 5.
- 6. Convocation Dates 2004

Spring Convocation 2004 will be held Tuesday, June 8, through Friday, June 11, and Fall Convocation 2004 will be held Thursday, October 21, and Friday, October 22.

7. 2001-2002 Winners of The Edward G. Pleva Award for Excellence in Teaching

The recipients of the Edward G. Pleva Award for Excellence in Teaching for 2001-2002 are:

- Donald Hair, Department of English, Faculty of Arts
- Stephen Hicock, Department of Earth Sciences, Faculty of Science
- John Howard, Department of Medicine, Faculty of Medicine & Dentistry
- Mitchell McInnes, Faculty of Law

Board of Governors APPENDIX VIII May 9, 2002 Page 6

8. 2001-2002 Winner of The UWO Award for Excellence in Teaching by Part-Time Faculty

The recipient of The UWO Award for Excellence in Teaching by Part-Time Faculty is:

• Mary Millard, Department of Statistical and Actuarial Science, Faculty of Science

9. 2001-2002 Winner of The Marilyn Robinson Award for Excellence in Teaching

The recipient of The Marilyn Robinson Award for Excellence in Teaching is:

Thomas Stavraky, Department of Physiology, Faculty of Medicine & Dentistry

10. **2002** Winners of the Hellmuth Prize for Achievement in Research

The winners of the 2002 Hellmuth Prize for Achievement in Research are Dr. Grant McFadden, Department of Microbiology and Immunology and Dr. Angela Esterhammer, Department of Modern Languages and Literatures.

11. **Announcements**

Brescia University College

Effective immediately, Brescia College will be known as Brescia University College.

Faculty of Information and Media Studies

C. Ross, Dean, July 1, 2002 - June 30, 2007

R. Babe, Jean Monty/BCE Chair in Media Studies, July 1, 2002 - June 30, 2007

Faculty of Medicine & Dentistry

P.L. Hooper, Chair, Dept. of Ophthalmology, July 1, 2002 - June 30, 2007

R.N. Rankin, Chair, Dept. of Diagnostic Radiology and Nuclear Medicine, July 1, 2002 - June 30, 2007 N.A.M. Paterson, Associate Dean - Policy and Planning, Hospital Liaison, effective July 1, 2001 - June 30, 2004.

Faculty of Social Science

B. Timney, Dean, July 1, 2002 - June 30, 2007

J.L. Granatstein, J.B. Smallman Memorial Chair in History, January 1, 2003 - April 30, 2003

Chairs and Professorships

G. Reid, The Bill and Anne Brock Family Professor in Child Health in the Departments of Psychology, Faculty of Social Science, and Family Medicine, Faculty of Medicine & Dentistry, effective July 1, 2001 - June 30, 2006

T.S. Sidhu, The Hydro One Chair in Power Systems Engineering, Department of Electrical and Computer Engineering, Faculty of Engineering, effective July 1, 2002.



The University of Western Ontario

Faculty of Medicine & Dentistry

CONDITIONS OF APPOINTMENT: PHYSICIANS APPOINTED IN CLINICAL DEPARTMENTS AND CLINICAL DIVISIONS OF BASIC SCIENCE DEPARTMENTS (1999)

(updated 2002)

Approved by Senate March 19, 1999
Approved by the Board of Governors March 25, 1999
Updates approved by Senate April 19, 2002
Updates approved by the Board of Governors (date)

TABLE OF CONTENTS

1	INTRODUCTION:	1
2	DEFINITIONS: 2.1 GENERAL: 2.2 APPOINTMENTS AND PROMOTIONS:	1
3	APPOINTMENTS AND REAPPOINTMENTS: 3.1 GENERAL: 3.2 ACADEMIC RANKS, CONTRACT TYPES AND TIME LIMITS: 3.3 PROCEDURES FOR INITIAL LIMITED TERM AND INITIAL CONTINUING APPOINTMENTS AND LIMITED TERM REAPPOINTMENTS: 3.4 CHANGES OF STREAM:	3 4 5
4	PROMOTION OR GRANTING OF A CONTINUING APPOINTMENT: 4.1 CRITERIA FOR FACULTY MEMBERS IN THE SENATE STREAM: 4.2 CRITERIA FOR FACULTY MEMBERS IN THE PROVOST STREAM: 4.3 PROCEDURES FOR PROMOTION AND GRANTING OF CONTINUING APPOINTMENTS: 4.4 ANNUAL REVIEW AND REVIEW OF CONTINUING APPOINTMENTS: 5.1	7 8 10
5	REVIEW COMMITTEES: COMPOSITION AND MANDATE: 1 5.1 CONFIDENTIALITY: 1 5.2 DEPARTMENT APPOINTMENTS AND PROMOTIONS COMMITTEE: 1 5.3 FACULTY APPOINTMENTS AND PROMOTIONS COMMITTEE: 1	12 13
6	APPEALS AGAINST PROMOTION, GRANTING OF CONTINUING APPOINTMENT, OR REVIEW OF CONTINUING APPOINTMENT: 6.1 APPEAL OF AN INITIAL NEGATIVE REVIEW BY THE DEPARTMENT: 6.2 APPEAL OF A DEPARTMENT LEVEL RECOMMENDATION: 6.3 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.4 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.5 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.6 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.7 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.8 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.9 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.1 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.1 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.2 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.3 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.4 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.5 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.6 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.7 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.8 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.9 APPEAL OF A FACULTY LEVEL RECOMMENDATION: 6.1 APPEAL A	15 16
7	APPEALS COMMITTEE: COMPOSITION AND MANDATE: 1 7.1 CONFIDENTIALITY 1 7.2 FACULTY COMMITTEE ON APPEALS: 1	17
8	LEAVES:	18
9	RESIGNATION, SUSPENSION AND DISMISSAL: 9.1 RESIGNATION: 9.2 SUSPENSION: 9.3 DISMISSAL: 1	18 19
10	GRIEVANCE: 2 10.1 GENERAL: 2 10.2 FACULTY GRIEVANCE COMMITTEE 2 10.3 UNIVERSITY GRIEVANCE COMMITTEE: 2	22 22

APPENDIX I: APPOINTMENTS PROCEDURES: MEMBERS OF ACADEMIC STAFF WITH LIMITED DUTIES (Physicians in the Faculty of Medicine & Dentistry)

1 INTRODUCTION:

- 1.1 These Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments set forth the academic ranks, terms of appointment and procedures for appointment, promotion and termination affecting Clinical Academics in clinical departments and clinical divisions of basic science departments in the Faculty of Medicine & Dentistry.
- 1.2 These faculty members make a significant contribution to the education of undergraduate and postgraduate students, residents and fellows, to research, and are major providers of care at the teaching hospitals and institutions affiliated with The University of Western Ontario. The University of Western Ontario recognizes the importance of these faculty members for excellence in teaching, research and scholarly activities, as well as for their contributions as role models, their health care leadership, and their general contributions in service or leadership which contribute to the teaching and research missions of the University. The Faculty of Medicine & Dentistry is committed to appointing faculty members who possess attributes of personal excellence and appropriate qualifications. All of these faculty members are responsible for discharging the various academic duties of the University and clinical duties of the teaching hospitals and affiliated institutions. Given the diverse academic and clinical priorities of the clinical departments and the clinical divisions of basic science departments, it is recognized that individual faculty members will develop different career pathways. These faculty members are valued and welcomed as full members of the professoriate of The University of Western Ontario.
- 1.3 The University of Western Ontario recognizes that a faculty member is free to criticize the University and to take controversial stands on public issues.
- 1.4 Appointments are made by the Board of Governors of The University of Western Ontario on the recommendation of the President and Vice-Chancellor of the University. The policies and procedures followed in recommending appointments to the Board of Governors are established and recommended by Senate.
- 1.5 Faculty members appointed under these Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments are required to comply with the policies, rules and regulations of the University.

2 **DEFINITIONS**:

2.1 GENERAL:

- 2.1.1 The "Act" shall mean <u>The University of Western Ontario Act</u> (1988) as the same may be amended from time to time.
- 2.1.2 The "University" shall mean The University of Western Ontario as established in the Act.
- 2.1.3 The **"Board of Governors"** or **"Board"** shall mean the Board of Governors of The University of Western Ontario as established in the Act.
- 2.1.4 The **"Senate"** shall mean the Senate of The University of Western Ontario as established in the Act.
- 2.1.5 The terms "academic staff", "Faculty", "Provost", "Dean", "Department", and related terms shall be interpreted in accordance with the Act and policies approved by the Senate and the Board of Governors.
- 2.1.6 **"Clinical divisions and departments"** are those whose members may have responsibility for patient care as well as the academic mission of the University.
- 2.1.7 "Department Committee" means the Department Appointments and Promotions Committee
- 2.1.8 **"Faculty Committee"** or **"Faculty Appointments and Promotions Committee"** means the Faculty of Medicine & Dentistry Appointments and Promotions Committee.

- 2.1.9 **"Affiliated Hospital or Institution"** shall refer to any hospital or other institution affiliated with The University of Western Ontario where the affiliation has been duly approved by the Board of Governors.
- 2.1.10 **"Clinician"** shall refer to an individual holding the degree of Medical Doctor or equivalent, licensed to practice medicine in Ontario by the College of Physicians and Surgeons of Ontario and actively engaged in the practice of medicine.

2.2 APPOINTMENTS AND PROMOTIONS:

- 2.2.1 "Appointment" means appointment to the academic staff of The University of Western Ontario. An appointment to the academic staff in accordance with these Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments does not imply or entail any remuneration. The remuneration, if any, to be received by the appointee in respect of his or her academic activities will be determined from time to time by the applicable departmental system for determining remuneration.
- 2.2.2 "Reappointment" means the renewal of a Limited Term Appointment, in accordance with the provisions set out in section 3. It may include a change of stream. A reappointment to the academic staff in accordance with these Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments does not imply or entail any remuneration. The remuneration, if any, to be received by the appointee in respect of his or her academic activities will be determined from time to time by the applicable departmental system for determining remuneration.
- 2.2.3 **"Promotion"** means a change in rank from Assistant Professor to Associate Professor or a change in rank from Associate Professor to Professor. Promotion from Assistant Professor to Associate Professor or promotion from Associate Professor, Limited Term to Professor may not occur without the granting of a Continuing Appointment.
- 2.2.4 **"Academic Job Description"** means the document which forms part of an appointment contract, specifying the designated stream and the general responsibilities of the appointee.
- 2.2.5 **A "Limited Term Appointment**" is for a specified period of time, up to three years, and may not extend beyond the effective date of retirement. Reappointment is neither promised nor guaranteed. Tenure does not apply to such appointments. A Limited Term Appointment is conditional upon registration in the College of Physicians and Surgeons of Ontario or any successor to that organization.
- 2.2.6 A **"Continuing Appointment"** is without an end date and may be held to retirement. A Continuing Appointment may only be held at the rank of Associate Professor or Professor. Tenure does not apply to such appointments. Continuance of a Continuing Appointment is conditional not only upon the satisfactory performance of the individual but also upon the factors identified in section 4.4 below. In addition, a Continuing Appointment is conditional upon registration in the College of Physicians and Surgeons of Ontario or any successor to that organization.
- 2.2.7 **"Practice Plan"** means the reporting mechanism to the department for clinical earnings or professional revenue, which may or may not include a mandatory contribution to a department or program academic enrichment fund, developed and accountable within the department, division or program and reviewed as acceptable by the University and Hospital at regular intervals.
- 2.2.8 "Letter of Appointment or Reappointment" means the letter to the prospective appointee signed by the Dean of the Faculty. This document specifies the academic contract type (Limited Term, Continuing Appointment), the starting and, if Limited Term, finishing dates of the appointment, the academic rank, Academic Job Description, stream, and any other applicable provisions or details. This letter forms part of the contracted relationship with the appointee.
- 2.2.9 **"Rank"** means the academic rank of Lecturer, Assistant Professor, Associate Professor, or Professor. The same nomenclature is used for all Clinical Academics, regardless of the "stream".
- 2.2.10 **"Stream"** means the designation within the Academic Job Description and Appointment which determines the set of academic criteria with which Clinical Academics must comply for promotion and/or the granting and maintenance of a Continuing Appointment. This designation recognizes that faculty members will develop different career paths and contribute to the academic objectives in different ways.

- 2.2.11 "Clinical Academic" is the appointment category of a clinician, who:
 - (i) is generally located in a teaching hospital, Family Medical Centre, or other clinical teaching unit;
 - (ii) contributes to the academic mission of the University with an on-going continuing commitment;
 - (iii) may be remunerated for academic contributions through a department or program merit-based reward system; and
 - (iv) is required to participate in a departmental practice plan.
- 2.2.12 "Adjunct" is the appointment category of a clinician who:
 - (i) is generally community-based, but may be located in an affiliated hospital or institution;
 - (ii) does not otherwise hold an academic appointment at the University (with the exception of other Adjunct or Limited-Duties Appointments):
 - (iii) performs specific limited responsibilities relating to teaching and/or research ancillary to the appointee's principal occupation;
 - (iv) generally receives no remuneration, although there may be a small stipend; and
 - (v) is not required to participate in a departmental practice plan.

The qualifications for the appointment emanate primarily from the individual's particular expertise, professional qualifications and/or experience within his or her occupational field. Adjunct appointments may be for up to five years and reappointment may be offered. Adjunct appointments may be made only as Adjunct Professor. There is no differentiated rank or stream associated with this category. Appointments are made under the document "Appointments Procedures: Members of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry)". (See Appendix I.)

- 2.2.13 "Honorary" is a category of appointment reserved generally for those who have attained academic or professional distinction in their field. Appointment as an Honorary Professor recognizes a mutually beneficial scholarly association between the University and the appointee and need not entail any teaching or other academic responsibilities on the part of the appointee. Appointments are made under the document "Appointments Procedures: Members of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry)." (See Appendix 1.)
- 2.2.14 "A Joint Appointment" is the appointment of a faculty member who:
 - (i) is appointed under this document as a Clinical Academic; and
 - (ii) has the equivalent of full-time duties distributed among two or more units on a continuing basis.

Joint appointments are made according to the policies and procedures for Joint Appointments within the University as approved by the Senate and the Board.

2.2.15 "Cross, Post-Retirement, and Visiting" are appointments made pursuant to the policies and procedures established by the University.

3 APPOINTMENTS AND REAPPOINTMENTS:

3.1 GENERAL:

- 3.1.1 No Limited Term Appointment made under this section shall be subject to termination either by the University or the appointee prior to the end of the specified period of the Limited Term, except and only in accordance with the provisions of these Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments.
- 3.1.2 No Continuing Appointment shall be subject to termination either by the University or the appointee, except and only in accordance with the provisions of these *Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments.*

3.1.3 Academic ranks shall be:

Lecturer Assistant Professor Associate Professor Professor

3.1.4 Appointments shall have a contract type of one of:

Limited Term (i.e., a specified period)
Continuing Appointment (i.e., without an end date)

- 3.1.5 Failure to renew a Limited Term Appointment does not constitute dismissal. Reappointment is neither promised nor guaranteed.
- 3.1.6 A Limited Term Appointment or Reappointment, or a Continuing Appointment, shall terminate immediately upon notice to the University of revocation of the appointee's registration in the College of Physicians and Surgeons of Ontario.
- 3.1.7 The granting of a Continuing Appointment is conditional upon a determination that the resources are sufficient to support the position. If there are insufficient resources, a Continuing Appointment will not be granted by the University. (See section 4.3.1.) If the University determines that the resources are sufficient to support the position, a decision to grant a Continuing Appointment shall be based on the record of performance achieved by the candidate in relation to the criteria established for the designated stream.
- 3.1.8 All appointments shall be to the stream designated in the Academic Job Description at the time of appointment or reappointment and the stream cannot be changed unless a new appointment is offered in accordance with section 3.4 below. The stream shall be one of:

Provost Stream Senate Stream

3.1.9 An academic review of a Continuing Appointment or a promotion shall be based on the record of performance achieved by the candidate in relation to the criteria established for the designated stream.

3.2 ACADEMIC RANKS, CONTRACT TYPES AND TIME LIMITS:

- 3.2.1 Initial appointment at the rank of Lecturer or Assistant Professor shall be for a Limited Term of up to three years.
- 3.2.2 The total continuous time spent at the rank of Lecturer, including reappointments, shall not normally exceed five years.
- 3.2.3 Reappointment may include a change of rank from Lecturer to Assistant Professor.
- 3.2.4 The total continuous time at the rank of Assistant Professor or at the rank of Assistant Professor combined with the rank of Lecturer shall not normally exceed seven years. The Department may recommend reappointment for a maximum of three years beyond the normal seven years. (See section 4.3.2.1.)
- 3.2.5 Initial appointment at the rank of Associate Professor may be on a Limited Term or Continuing Appointment basis:
- 3.2.5.1 An initial Limited Term Appointment at the rank of Associate Professor shall be for up to three years. The total continuous time spent on a Limited Term basis at the rank of Associate Professor at The University of Western Ontario, including reappointments, shall not exceed five years.
- 3.2.5.2 An initial appointment at the rank of Associate Professor with a Continuing Appointment shall follow the procedures set out in section 3.3 and the relevant provisions that govern granting of a Continuing Appointment, including section 4.
- 3.2.6 An initial appointment at the rank of Professor shall be made as a Continuing Appointment and shall follow the

procedures set out in section 3.3 and the relevant provisions that govern granting of a Continuing Appointment, including section 4.

3.3 PROCEDURES FOR INITIAL LIMITED TERM AND INITIAL CONTINUING APPOINTMENTS AND LIMITED TERM REAPPOINTMENTS:

- 3.3.1 The filling of open positions shall follow relevant University policies and procedures.
- 3.3.2 The Department Committee shall consider all open academic positions in the Department and the eligibility for reappointment of all members with Limited Term Appointments whose current term of appointment is ending that year.
- 3.3.2.1 If the initial appointment is a Continuing Appointment at the rank of Associate Professor or Professor, the Department Chair shall ensure that an appropriate file is established in accordance with section 4.3.7. containing the curriculum vitae of the recommended candidate and any letters of reference for the recommended candidate used by the Department Committee along with any other supporting documentation on the recommended candidate's performance (e.g., teaching) which may be available. (See also sections 3.2.5.2 and 3.2.6.)
- 3.3.3 The Department Chair and the Department Committee shall each make separate recommendations concerning the academic rank, the contract type, the stream, and the term of the appointment or reappointment.
- 3.3.3.1 If the recommendations of both the Department Chair and Department Committee are positive, both recommendations shall be sent to the Dean by the Department Chair within a reasonable period of time.
- 3.3.3.2 If the recommendations of both the Department Chair and Department Committee are negative, the recommendations will not be forwarded to the Dean and no appointment or reappointment will be offered. There is no right of appeal of this decision. The Department Chair shall inform the individual in writing of the decision.
- 3.3.3.3 If the recommendation of the Department Chair is at variance with that of the Department Committee, the Department Chair shall inform the Committee and both recommendations shall be sent to the Dean by the Department Chair within a reasonable period of time. Under these circumstances, the Department Chair shall await the instruction of the Dean before taking any further action.
- 3.3.4 If a recommendation is sent to the Dean pursuant to sections 3.3.3.1 or 3.3.3.3 for an initial Limited Term Appointment or a Limited Term Reappointment at the rank of Lecturer, Assistant Professor or Associate Professor, the Dean shall make a recommendation.
- 3.3.4.1 If the recommendations of both the Department Chair and Department Committee are positive, and the Dean makes a positive recommendation, all of the recommendations shall be sent to the Provost by the Dean within a reasonable period of time.
- 3.3.4.2 If the recommendation of the Dean is at variance with that of the Department Chair and/or Department Committee, the Dean shall request a consultation with the Department Chair and Department Committee. The Dean shall inform the Department Chair and Department Committee of the Dean's final recommendation and forward that recommendation together with the recommendations of the Department Chair and the Department Committee to the Provost.
- 3.3.4.3 The Provost shall review the recommendations and, if there are any concerns, shall consult with the Dean before making a final decision. The decision of the Provost is final and not subject to appeal.
- 3.3.5 If a recommendation is sent to the Dean pursuant to sections 3.3.3.1 or 3.3.3.3 for an initial Continuing Appointment at the rank of Associate Professor or Professor, the Dean and Faculty Committee shall each make separate recommendations.
- 3.3.5.1 If the recommendations of the Dean and/or Faculty Committee are positive, the Dean shall forward both recommendations to the Provost. If the recommendation of the Dean is at variance with that of the Faculty

Committee, the Dean shall inform the Committee.

- 3.3.5.2 The Provost shall review the recommendations for appointment as Associate Professor, Continuing Appointment or Professor and, if there are any concerns, shall consult with the Dean and Faculty Committee before making a final decision. There is no right of appeal of this decision.
- 3.3.5.3 If the recommendations of both the Dean and Faculty Committee are negative, the recommendations will not be forwarded to the Provost and no appointment/reappointment will be offered. There is no right of appeal of this decision.
- 3.3.6 When the required approvals have been obtained, the Department Chair shall undertake negotiations with the candidate in consultation with the Dean.
- 3.3.7 The Dean shall inform the individual in writing of the terms and conditions of the appointment or reappointment. These terms and conditions shall not conflict with these *Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions in Basic Science Departments.*

3.4 CHANGES OF STREAM:

- 3.4.1 In a case where a faculty member with a Limited Term Appointment or a Continuing Appointment requests a new appointment in a different stream, consideration shall be given by the Department Chair to such an appointment with the same rank and contract type, based on the needs of the University and the relevant affiliated hospital or institution. If the Department Chair decides that the request for a new appointment in a different stream should be given consideration by the University, the relevant procedures for approval of initial appointments set out in sections 3.3.2.1 to 3.3.7 must be followed, except that the appropriate file described in section 3.3.2.1 shall be established in accordance with section 4.3.7.
- 3.4.1.1 Should the new appointment be approved, this shall replace the existing appointment. For purposes of section 3.2 governing time limits at a rank or ranks, all continuous time in both streams shall be included.
- 3.4.1.2 Should the new appointment not be approved:
- 3.4.1.2.1 An existing Limited Term Appointment shall remain in effect until the end of the term subject to sections 3.1.6 and 9.
- 3.4.1.2.2 An existing Continuing Appointment shall remain in effect, subject to sections 3.1.6, 4.4 and 9.
- 3.4.1.2.3 There is no right of appeal of a negative decision regarding the new appointment.

4 PROMOTION OR GRANTING OF A CONTINUING APPOINTMENT:

4.1 CRITERIA FOR FACULTY MEMBERS IN THE SENATE STREAM:

- 4.1.1 Each candidate for promotion and/or the granting of a Continuing Appointment is expected to establish a record of performance in teaching, in research and in general contributions.
- 4.1.2 The record of performance in both teaching and research must be sufficiently strong to warrant promotion and/or the granting of a Continuing Appointment by the University.
- 4.1.3 The performance shall be considered with reference to the national and international standards within the candidate's discipline.
- 4.1.4 Normally, the significance accorded to the first two criteria, teaching and research, is approximately equal and

individually each must be accorded greater significance than the third criterion, general contributions.

- While a candidate must have achieved a satisfactory record of performance in general contributions, meritorious performance in this area shall not compensate for an insufficiently strong record of performance in both teaching and research. However, an unsatisfactory record of performance in general contributions may be an important factor in the delaying or denial of promotion and/or the denial of a Continuing Appointment.
- 4.1.6 The Faculty Committee may accord substantially greater significance or weight either to teaching or to research in its recommendation. If the Faculty Committee accords substantially greater significance to either teaching or research, the candidate will be expected to have achieved an outstanding record of performance in that area. A candidate who has achieved an outstanding record of performance in either teaching or research must also establish a reasonable record of performance in the other area. If substantially greater significance is accorded to either teaching or research, the significance accorded to the other area still must be greater than that accorded to the area of general contributions. If the Faculty Committee has accorded substantially greater significance to either teaching or research, it must include with its recommendation its reasons for concluding that the candidate has established an outstanding record of performance in one area and a reasonable record of performance in the other area.
- 4.1.7 The rank of Associate Professor shall be a career rank.
- 4.1.8 The rank of Professor is the highest rank that the University can bestow, and its conferral shall be in recognition of high achievement in teaching and research. A candidate for appointment or promotion to the rank of Professor is expected to have established a record of performance in teaching and in research that is substantially greater than the record of performance expected of an Associate Professor. Although high achievement over a sustained period shall normally be expected of a successful candidate for the rank of Professor, length of service alone shall not be a criterion for promotion. While the recommendation for appointment or promotion to the rank of Professor shall be based primarily on the record of performance in teaching and in research, a candidate must also have established a significant record of general contributions.
- 4.1.9 At the time of the consideration of a candidate, evidence shall be provided to the appropriate committees that the candidate has established a record of performance consistent with the requirements above and in accordance with the following criteria:
- 4.1.9.1 <u>Performance in Teaching and Associated Activities</u>

This includes teaching and evaluating undergraduate students, graduate students, postgraduate students, and practicing health care professionals.

Performance in teaching and associated activities may also include but is not limited to:

- (i) Initiatives in course design and curriculum development;
- (ii) Initiatives designed to improve clinical teaching;
- (iii) Development of effective and innovative teaching resources, including computer courseware, and keeping up with technological developments where relevant;
- (iv) Administration in relation to effective educational planning and policy making; and
- (v) Instructional development -- activities intended to assist the faculty members to improve their teaching.

4.1.9.2 Performance in Research and Scholarly Activities

Performance in this category includes the following:

- (i) Publication of the results of original research and clinical investigations;
- (ii) Contribution to the advancement of knowledge through publication of new concepts or techniques, invention of medical apparatus, description of illnesses or critical review of published work;
- (iii) Presentations at professional and scientific meetings;
- (iv) Visiting professorships to other universities;

NOTE: Publications and presentations related to research and scholarly activities in teaching and education may also be included.

In evaluating this performance, the written opinion of at least three independent experts in the candidate's area of specialization who are not members of this University shall be sought.

In evaluating research, creativity and quality shall be assessed as well as industry.

4.1.9.3 <u>General Contributions in Service or Leadership Within the University, Profession, and/or the Community Which Contribute to the Teaching and Research Missions of the University:</u>

Performance in this category may include but is not limited to the following:

- (i) Major administrative responsibilities;
- (ii) Editorial duties;
- (iii) Memberships on boards;
- (iv) Leadership roles in professional organizations; and
- (v) Leadership and participation in University committees.

4.2 CRITERIA FOR FACULTY MEMBERS IN THE PROVOST STREAM:

- 4.2.1 Each candidate for promotion and/or the granting of a Continuing Appointment is expected to establish a record of performance demonstrating that the candidate is creating and disseminating knowledge and providing leadership. The candidate is expected to establish a record of performance in at least one of teaching or research and at least one of role model, health care leadership or general contributions.
- 4.2.2 The record of performance must be sufficiently strong to warrant promotion and/or the granting of a Continuing Appointment by the University.
- 4.2.3 The performance shall be considered with reference to the national and international standards within the candidate's discipline.
- 4.2.4 The rank of Associate Professor shall be a career rank.
- 4.2.5 The rank of Professor is the highest rank that the University can bestow, and its conferral shall be in recognition of high achievement. A candidate for appointment or promotion to the rank of Professor is expected to have established a record of performance that is substantially greater than the record of performance expected of an Associate Professor. Although high achievement over a sustained period shall normally be expected of a successful candidate for the rank of Professor, length of service alone shall not be a criterion for promotion.
- 4.2.6 At the time of the consideration of a candidate, evidence shall be provided to the appropriate committees and to the Provost that the candidate has established a record of performance consistent with the requirements above and in accordance with the following criteria:
- 4.2.6.1 <u>Performance in Teaching and Associated Activities</u>:

This includes teaching and evaluating undergraduate students, graduate students, postgraduate students, and practising health care professionals.

Performance in teaching and associated activities may also include but is not limited to:

- (i) Initiatives in course design and curriculum development;
- (ii) Initiatives designed to improve clinical teaching;
- (iii) Development of effective and innovative teaching resources, including computer courseware, and keeping up with technological developments where relevant;
- (iv) Administration in relation to effective educational planning and policy making; and
- (v) Instructional development -- activities intended to assist the faculty members to improve their teaching.

4.2.6.2 Performance in Research and Scholarly Activities:

Performance in this category includes the following:

- (i) Publication of the results of original research and/or clinical investigations;
- (ii) Contribution to the advancement of knowledge through publication of new concepts or techniques, invention of medical apparatus, description of illnesses or critical review of published work;

- (iii) Presentations at professional and scientific meetings;
- (iv) Visiting professorships to other universities;

NOTE: Publications and presentations related to research and scholarly activities in teaching and education may also be included.

In evaluating this performance, the written opinion of at least three independent experts in the candidate's area of specialization who are not members of this University shall be sought. In evaluating research, creativity and quality shall be assessed as well as industry.

4.2.6.3 Role Model:

Clinical or laboratory practice which contributes to the academic missions of the University. Performance in this category may include but is not limited to the following:

- (i) Expert/excellent clinical or laboratory practitioner dedicated to provision of quality patient care;
- (ii) Development of innovative techniques;
- (iii) Introduction of new techniques to the University medical community;
- (iv) Improvements in clinical or laboratory practice; and
- (v) Recognized regular contributor of patients and data to clinical trials.

4.2.6.4 Health Care Leadership:

Performance in this category may include but is not limited to the following:

- (i) Health care administration which involves policy development and implementation of change;
- (ii) Participation in programs, projects, committees, or consulting roles which have an impact on health care; and
- (iii) Health care communication/education of patients and the community.
- 4.2.6.5 <u>General Contributions in Service or Leadership Within the University, Profession, and/or the Community</u> Which Contribute to the Teaching and Research Missions of the University:

Performance in this category may include but is not limited to the following:

- (i) Major administrative responsibilities;
- (ii) Editorial duties;
- (iii) Memberships on boards;
- (iv) Leadership roles in professional organizations; and
- (v) Leadership and participation in University committees.

4.3 PROCEDURES FOR PROMOTION AND GRANTING OF CONTINUING APPOINTMENTS:

- 4.3.1 The following procedures shall apply only where the resources and needs of the University (including departmental, program, and budgetary considerations) are such that the granting of a continuing appointment to a clinical academic is feasible; otherwise, the limited term appointment will terminate at the end of the term and the following procedures shall not apply.
- 4.3.2 Subject to section 4.3.1 above, all Clinical Academics at the rank of Assistant Professor shall be considered for promotion to Associate Professor, Continuing Appointment, within seven years from the time of first appointment as a Clinical Academic.
- 4.3.2.1 Clinical Academics at the rank of Assistant Professor who are not recommended for promotion to Associate Professor, Continuing Appointment, in the seventh year may be offered a Limited Term reappointment for a maximum of three years but, if so, must be given consideration for promotion to Associate Professor, Continuing Appointment, in the tenth year, subject to section 4.3.1 above.
- 4.3.3 Subject to section 4.3.1 above, Clinical Academics appointed at the rank of Associate Professor, Limited Term shall be considered for granting of a Continuing Appointment within five years from the time of first appointment as a Clinical Academic.
- 4.3.4 Failure to achieve promotion to Associate Professor, Continuing Appointment, or the granting of Continuing

Appointment in the case of an Associate Professor, Limited Term, within the maximum time limits set out in section 3.2, shall result in no further reappointment as a Clinical Academic.

- 4.3.5 The Department Committee shall annually review:
- 4.3.5.1 all Clinical Academics at the rank of Assistant Professor who are entering the seventh or tenth year from their initial year of appointment, for promotion to Associate Professor, Continuing Appointment;
- 4.3.5.2 all Clinical Academics at the rank of Associate Professor, Limited Term, who are entering their fifth year from their initial year of appointment, for granting of a Continuing Appointment;
- 4.3.6 In addition, the Department Committee shall annually consider whether any other Clinical Academics at the rank of Assistant Professor or Associate Professor, should be reviewed for promotion or the granting of a Continuing Appointment.
- Following this consideration, the Department Chair shall advise those faculty members whose cases, in the Committee's opinion, warrant review. In cases where a faculty member in the Senate stream has requested consideration for promotion to Professor, Senate stream, but, in the Department Committee's opinion, a review is not warranted, the Department Chair shall advise the faculty member of this decision and provide the faculty member with a statement of reasons. In this situation, the faculty member may appeal under section 6.1.
- 4.3.7 The Department Chair shall ensure that an appropriate file is established for Clinical Academics at the rank of Assistant Professor entering the seventh or tenth year from initial appointment, for Clinical Academics at the rank of Associate Professor, Limited Term entering their fifth year from initial appointment, and in other cases where, in the Committee's opinion, a review is warranted. The file shall include:
 - (i) A current curriculum vitae;
 - (ii) Any written submissions from the candidate relevant to the case;
 - (iii) Evaluations of teaching performance from students and, where possible, from faculty;
 - (iv) Evaluation of research and scholarly activity by confidential reports from at least three independent referees external to the University who are experts in the candidate's discipline. [This will not always be necessary for candidates in the Provost stream who are not involved in research.];
 - (v) Evaluation of performance in other relevant criteria. This evaluation shall include reports of peers; and
 - (vi) Recommendations from each of the Department Chair and Department Committee.

Upon request to the Chair of the relevant committee, a faculty member may review and photocopy the file. However, in accordance with the University's policy of maintaining confidentiality, prior to the file being seen by the candidate, all reports of external referees, students and peers shall have all traces of their place of origin and authorship removed.

- 4.3.8 The Department Chair and the Department Committee shall review the file and each shall make a separate recommendation. The candidate shall be informed of these recommendations by October 1st:
- 4.3.8.1 If the recommendations of the Department Chair and/or Department Committee are positive, both recommendations shall be forwarded to the Dean by the Department Chair with the file by October 1st. If the recommendation of the Department Chair is at variance with that of the Department Committee, the Department Chair shall inform the Committee.
- 4.3.8.2 If the recommendations of both the Department Chair and Department Committee are negative, the recommendations shall not be forwarded to the Dean. A statement of reasons for the decision shall be given to the candidate by October 1st. The candidate shall have the right to appeal this decision under section 6.2.
- 4.3.9 The Dean and the Faculty Committee shall review the file and recommendations forwarded to the Dean pursuant to section 4.3.8.1, and each shall make a separate recommendation. These recommendations shall be sent in writing to the candidate and to the Department Chair by December 1st.
- 4.3.9.1 If the recommendations of the Dean and/or Faculty Committee are positive, the Dean shall forward both recommendations to the Provost. If the recommendation of the Dean is at variance with that of the Faculty Committee, the Dean shall inform the Committee.

- 4.3.9.2 If the recommendations of both the Dean and Faculty Committee are negative, the recommendations will not be forwarded to the Provost. A statement of reasons for the decision shall be given to the candidate. The candidate shall have the right to appeal this decision under section 6.3.
- 4.3.10 If the Provost receives positive recommendations from the Dean and/or Faculty Committee pursuant to section 4.3.9.1.
- 4.3.10.1 the Provost shall review the recommendations for promotion to Associate Professor, Continuing Appointment, Provost stream, recommendations for the granting of a Continuing Appointment, Provost stream, in the case of an Associate Professor, Limited Term, and recommendations for promotion to Professor, Provost stream, and, if there are any concerns, shall consult with the Dean, representatives of the Faculty Committee and the candidate before making a final decision. There is no right of appeal of this decision.
- the Provost shall review the recommendations for promotion to Associate Professor, Continuing Appointment, Senate stream, recommendations for the granting of a Continuing Appointment, Senate stream, in the case of an Associate Professor, Limited Term, and recommendations for promotion to Professor, Senate stream, and if there are any concerns, shall consult with the Dean, representatives of the Faculty Committee and the candidate before making a final decision. The Provost's decision may be grieved to a University Grievance Committee under section 10.3.

4.4 ANNUAL REVIEW AND REVIEW OF CONTINUING APPOINTMENTS:

- 4.4.1 Each faculty member must maintain a minimum standard of performance in accordance with the criteria set out in sections 4.1 or 4.2, as applicable. The Department Chair is responsible, on an annual basis, for reviewing the performance of faculty members in the Department. This review will involve an evaluation of the faculty member's performance relative to the Departmental needs, objectives and priorities. The Dean is responsible, on an annual basis, for reviewing summary information from each department on all faculty members. The Dean may seek such advice and information as he or she deems necessary in this regard.
- 4.4.2 A formal review of the record of performance of a faculty member with a Continuing Appointment shall be initiated by the Department Chair and/or the Dean as follows:
- 4.4.2.1 If the Department Chair determines that a record of performance fails to meet minimum standards, the Department Chair shall inform the faculty member in writing of the area(s) which require improvement. The faculty member may respond in writing, providing additional information. All correspondence shall be copied to the Dean.
- 4.4.2.2 If the Dean determines independently that the record of performance has not been maintained, the faculty member and Department Chair will be informed in writing of the area(s) which require improvement. The faculty member and Department Chair may respond in writing, providing additional information.
- 4.4.2.3 If the Department Chair and/or Dean determine that a faculty member has failed to maintain a minimum standard of performance for two successive years, a review of the Continuing Appointment shall be initiated by the Chair upon written notification to the faculty member. A file shall be established as described in section 4.3.7.
- 4.4.3 The Department Chair and Department Committee shall review the file within two months of the date of the written notification to the faculty member. Each shall make a separate recommendation regarding whether the minimum standard of performance has been maintained. These recommendations shall be sent in writing to the faculty member within two weeks of the decision.
- 4.4.3.1 The Department Chair shall forward both recommendations to the Dean with the file. If the recommendation of the Department Chair is at variance with that of the Department Committee, the Department Chair shall inform the Committee. In the case of a negative recommendation by the Department Chair and/or the Department Committee, a statement of reasons for the decision shall be included with the recommendation and given to the faculty member.
- 4.4.4 The Dean and the Faculty Committee shall review the file and recommendations forwarded to the Dean pursuant to section 4.4.3.1 and each shall make a separate decision regarding whether the minimum standard of

performance has been maintained. These decisions shall be sent in writing to the Provost, to the candidate, and to the Department Chair:

- 4.4.4.1 If the decisions of the Dean and/or Faculty Committee are positive, the appointment shall continue, subject to future reviews under section 4.4. The Dean shall forward both decisions to the Provost. If the decision of the Dean is at variance with that of the Faculty Committee, the Dean shall inform the Committee.
- 4.4.4.2 If the decisions of both the Dean and Faculty Committee are negative the appointment shall terminate with three months' notice of termination to the faculty member. A statement of reasons for the decisions shall be given to the faculty member and a copy shall be forwarded to the Provost. The faculty member shall have the right to appeal the decisions under section 6.3.

5 REVIEW COMMITTEES: COMPOSITION AND MANDATE:

5.1 CONFIDENTIALITY:

- 5.1.1 All members of any committee mandated under these *Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments* shall respect the confidentiality of the committee's deliberations, consultations and any other relevant committee activities or proceedings.
- 5.1.2 This constraint shall also apply to all other individuals who may from time to time be required to appear before or otherwise be involved in the <u>in camera</u> proceedings of any such committee in the capacity of consultant, counsel or academic colleague, witness or party.

5.2 DEPARTMENT APPOINTMENTS AND PROMOTIONS COMMITTEE:

- 5.2.1 Each basic science department in the Faculty of Medicine & Dentistry with a clinical division shall have an Appointments and Promotions Committee. The structure of the Committee shall be determined by the department but shall include:
 - (i) the Department Chair who shall Chair the Committee;
 - (ii) at least two full-time tenured members of the Department elected by the full-time members of the Department;
 - (iii) one or more Clinical Academics with Continuing Appointments at the rank of Associate Professor or Professor from the clinical division elected by the Clinical Academics in the Department;
 - (iv) in addition, where the Department so chooses, untenured members of Faculty and/or students, all of whom shall be elected by the Department;
 - (v) where a Department is unable to meet the requirements outlined in either (ii) or (iii) above, Faculty Council shall elect the members required under either (ii) or (iii) above with the proviso that at least one-half of the members of the Appointments Committee shall be elected from among the faculty members in the Department.
- 5.2.2 Each clinical department in the Faculty of Medicine & Dentistry shall have an Appointments and Promotions Committee. The structure of the Committee shall be determined by the department but shall include:
 - (i) the Department Chair who shall Chair the Committee;
 - (ii) at least two Clinical Academic members of the Department with Continuing Appointments at the rank of Associate Professor or Professor elected by the Clinical Academics in the Department:
 - (a) if the Department has more than one Clinical Academic at the rank of Associate Professor or Professor, Senate stream: at least one Clinical Academic, Senate stream, at the rank of Associate Professor or Professor; and
 - (b) if the Department has more than one Clinical Academic at the rank of Associate Professor or Professor.

Provost stream: at least one Clinical Academic, Provost stream, at the rank of Associate Professor or Professor.

- (iii) where a department has fewer than two Clinical Academics with Continuing Appointments at the rank of Associate Professor or Professor, either stream (excluding the Department Chair), the Committee members shall be elected by Faculty Council. In such a case, at least one half of the members of the Department Committee shall be elected from among the faculty members in the Department.
- 5.2.3 In the event that an elected member of the Department Committee is to be considered for promotion and/or reappointment, the member shall retire from the Committee.
- 5.2.4 The membership of the Department Committee shall be reported annually to the Dean of the Faculty of Medicine & Dentistry.
- 5.2.5 The Committee shall be convened by the Department Chair.
- 5.2.6 Should any member of the Committee, including the Department Chair, have a conflict of interest, that person shall withdraw from consideration of the relevant case or cases. Should the Department Chair have a conflict of interest, he or she shall refrain from making a separate recommendation in the relevant case(s).
- 5.2.7 The Department Committee shall consider all open academic positions in the Department, all proposed appointments involving a change of stream, the eligibility for reappointment of all Department members with Limited Term Appointments whose current term of appointment is ending that year, all promotions and granting of Continuing Appointment for Departmental members, and shall review Continuing Appointments as provided in section 4.4.3.
- 5.2.7.1 In considering appointment, reappointment or promotion, the Committee shall have regard for the time limits set out in section 3.2.
- 5.2.7.2 In considering promotion, granting of a Continuing Appointment, and reviewing of a Continuing Appointment, the Committee shall have regard for the criteria set out in sections 4.1 and 4.2 and the fact that a Faculty member is free to criticize the University and to take controversial stands on public issues.

5.3 FACULTY APPOINTMENTS AND PROMOTIONS COMMITTEE:

- 5.3.1 The Faculty of Medicine & Dentistry shall have a Faculty Committee, the composition of which shall be:
 - (i) the Dean (or designated Associate Dean), who shall be Chair;
 - (ii) two Clinical Academics at the rank of Professor, elected by Faculty Council, one of whom shall be Senate stream and one of whom shall be Provost stream;
 - (iii) one full-time tenured Professor from the basic science departments, elected by Faculty Council;
 - (v) one full-time tenured Professor from the Faculty of Science, elected by the Faculty of Science Council.
- 5.3.2 No more than one member may be from any one department.
- 5.3.3 A member of a Department Committee may serve simultaneously on the Faculty Committee, subject to conflict of interest rules.
- 5.3.4 The membership of the Faculty Committee shall be reported annually to Senate through the Secretary of Senate.
- 5.3.5 Quorum shall consist of two-thirds of the members and must include the Dean (or designated Associate Dean) unless the Dean (or designated Associate Dean) has withdrawn in which case quorum shall consist of two-thirds of the remaining members.
- 5.3.6 Should any member of the Committee have a conflict of interest, that person shall withdraw from consideration of the relevant case or cases. Should the Dean (or designated Associate Dean) have a conflict of interest, the Dean (or designated Associate Dean) shall also refrain from making a separate recommendation in the relevant case(s).
- 5.3.7 The mandate of the Faculty Committee is as follows:

- 5.3.7.1 The Committee shall consider all positive recommendations for initial appointments at the rank of Associate Professor, Continuing Appointment, or Professor. Such recommendations may emanate from the Department Chair or Department Committee or from a Selection Committee in accordance with the policies and procedures established by the University.
- 5.3.7.2 In cases where the Department Chair and/or the Department Committee have made positive recommendations relating to,
 - (i) granting of a Continuing Appointment combined with promotion to the rank of Associate Professor;
 - (ii) granting of a Continuing Appointment to a faculty member with a Limited Term Appointment at the rank of Associate Professor; and
 - (iii) promotion to Professor, Continuing Appointment,

the Committee shall review all recommendations from the Department Chair and the Department Committee so as to preserve a similarity of standards from department to department.

- 5.3.7.3 The Committee shall consider all recommendations for appointments of faculty members at the rank of Associate Professor, Continuing Appointment or Professor, Continuing Appointment, involving a change of stream in cases where the Department Chair and/or the Department Committee have made positive recommendations.
- 5.3.7.4 The Committee shall consider all recommendations from a Department Chair and Department Committee arising from reviews of Continuing Appointments initiated by the Department Chair or the Dean.
- In cases where there has been a positive recommendation by the Department Chair or the Department Committee, but the Dean or Faculty Committee have concerns about making a positive recommendation, the Faculty Committee shall hold a consultation with the Department Chair and Department Committee prior to making a final recommendation.
- 5.3.7.6 The Faculty Committee shall hear appeals against negative decisions of the Department Chair and Department Committee concerning promotion, and/or the granting of a Continuing Appointment (other than initial Continuing Appointments or new Continuing Appointments involving a change of stream), where neither the Department Chair nor the Department Committee has made a positive recommendation.

NOTE: Appeals may be initiated only after formal consultation between the candidate and the Department Committee has taken place in accordance with Section 6.2.

6 APPEALS AGAINST PROMOTION, GRANTING OF CONTINUING APPOINTMENT, OR REVIEW OF CONTINUING APPOINTMENT:

6.1 APPEAL OF AN INITIAL NEGATIVE REVIEW BY THE DEPARTMENT:

- 6.1.1 A Senate Stream faculty member at the rank of Associate Professor has the right to consult with the Department Chair and the Department Committee when the Department Committee has decided his or her case for promotion to Professor, Senate Stream, will not receive further consideration that year under section 4.3.6.1.
- 6.1.2 Within one week from the date upon which the letter giving notification of the Department Committee's decision and statement of reasons was sent to the faculty member, the faculty member may write to the Department Chair requesting a meeting with the Department Chair and Department Committee, submitting all further documentation that he or she deems relevant to the case.
- 6.1.3 The meeting shall be held within one week of the Department Chair's receipt of the written request from the faculty member.
- 6.1.4 Following the meeting, the Department Chair and Department Committee shall confirm or revise the initial decision that the case does not warrant further consideration for promotion. If the Department Committee decides that the faculty member's record does warrant further consideration, the faculty member shall be notified within one week

- of the consultation that a promotion file is being established and the procedures set out in 4.3.7 to 4.3.10.2 shall be followed.
- 6.1.5 If the Department Committee confirms the negative outcome of the initial review, the appellant must be informed of the decision, with a written statement of reasons, within one week of the consultation. The faculty member may appeal this decision to the Faculty Committee under 6.1.6.
- 6.1.6 An appeal to the Faculty Committee shall be made in writing, stating all grounds of appeal, within four weeks from the date upon which the letter giving notification of the Department Chair and Department Committee's decision was sent to the faculty member.
- 6.1.7 The Faculty Committee shall meet to hear the appeal within four weeks of receipt of the appeal. The faculty member and the Department Chair, with at least one representative from the Department Committee, shall be invited to attend. The faculty member may be accompanied by an academic colleague.
- 6.1.8 Following the meeting, if the Faculty Committee agrees that the appellant's case for promotion warrants further consideration, the Department will be instructed to establish a promotion file and the procedures set out in 4.3.7 to 4.3.10.2 shall be followed.
- 6.1.9 If the Faculty Committee upholds the decision of the Department Chair and Department Committee that further consideration of the faculty member's case for promotion is not warranted, no promotion file will be established for the appellant and no further consideration will be given to the case that year. The appellant must be informed of the Faculty's negative decision, with a written statement of reasons, within one week of the meeting. The Faculty Committee's decision shall be grievable to a University Grievance Committee under 10.3..
- 6.1.10 If the University Grievance Committee agrees that the appellant's case for promotion warrants further consideration, the Department, through the Dean of the Faculty, will be instructed to establish a promotion file and the procedures set out in 4.3.7 to 4.3.10.2 shall be followed.
- 6.1.11 If the University Grievance Committee upholds the decision of the Faculty Committee that further consideration of the faculty member's case for promotion is unwarranted, no promotion file will be established for the appellant and no further consideration will be given to the case that year.
- 6.1.12 The decision of the University Grievance Committee concerning the Faculty's decision not to give further consideration to the promotion of a faculty member in a given year shall be final.

6.2 APPEAL OF A DEPARTMENT LEVEL RECOMMENDATION:

- 6.2.1 A faculty member has the right to consult with the Department Chair and the Department Committee when both the Department Chair and Department Committee have made a negative recommendation regarding promotion and/or the granting of a Continuing Appointment (other than negative decisions resulting from consideration of an initial Continuing Appointment or a new Continuing Appointment involving a change of stream).
- 6.2.2 Within one week from the date upon which the letter giving notification of the decision and statement of reasons was sent to the faculty member, the faculty member may write to the Department Chair requesting a meeting with the Department Chair and Department Committee, submitting all further documentation that he or she deems relevant to the case.
- 6.2.3 The meeting shall be held within four weeks of the Department Chair's receipt of the written request from the faculty member. The faculty member may be accompanied by an academic colleague.
- 6.2.4 Following the meeting, the Department Chair and Department Committee shall each make a final decision on the case. If the final recommendations of both the Department Chair and Department Committee are negative, the Department Chair shall provide the faculty member with a written statement of reasons within one week of the meeting.
- 6.2.5 If the final decisions of both the Department Chair and Department Committee are negative, the faculty member may appeal to the Dean and Faculty Committee. An appeal to the Faculty Committee shall be made in writing, stating all grounds of appeal, within four weeks from the date from which the letter giving notification of the final

recommendation was sent to the faculty member.

- 6.2.6 The Faculty Committee shall meet to hear the appeal within four weeks of receipt of the appeal. The faculty member and the Department Chair, with at least one representative from the Department Committee, shall be invited to attend. The faculty member may be accompanied by an academic colleague.
- 6.2.7 Following the meeting, the Dean and Faculty Committee shall each make a final recommendation on the case and shall provide a written statement of reasons to the faculty member within one week of the meeting. If the recommendations of both the Dean and the Faculty Committee are negative, the faculty member may appeal under section 6.3.3 1 or 6.3.3.2, as appropriate. If the recommendations of the Dean and/or Faculty Committee are positive, the Dean shall follow the procedures set out in section 4.3.9.1.
- 6.2.8 A negative decision of a Department Chair and/or Department Committee resulting from consideration of an initial appointment or reappointment (renewal of a Limited Term Appointment) may not be appealed.
- 6.2.9 A negative decision of a Department Chair and/or Department Committee resulting from consideration of a new appointment involving a change of stream may not be appealed; however, where there is an objection on procedural grounds, see section 10.1.2.
- 6.2.10 A negative decision of a Department Chair and/or Department Committee resulting from review of a Continuing Appointment is automatically reviewed when the file is forwarded to the Dean and Faculty Committee. (See section 4.4.4.) If the final recommendation of both the Dean and Faculty Committee are negative, see section 6.3.3.

6.3 APPEAL OF A FACULTY LEVEL RECOMMENDATION:

- 6.3.1 A faculty member has the right to consult with the Dean and the Faculty Committee when both the Dean and the Faculty Committee have made negative recommendations resulting from consideration of promotion and/or the granting of Continuing Appointment (other than negative decisions subsequent to an appeal to the Faculty Committee under section 6.2 above and negative decisions resulting from consideration of an initial Continuing Appointment or a new Continuing Appointment involving a change of stream).
- 6.3.2 A faculty member has the right to consult with the Dean and the Faculty Committee when the Dean and the Faculty Committee have made negative decisions concerning a review of a Continuing Appointment.
- 6.3.2.1 Within one week from the date upon which the letter giving notification of the decision and statement of reasons was sent to the faculty member, the faculty member may write to the Dean requesting a meeting with the Dean and the Faculty Committee, submitting all further documentation that he or she deems relevant to the case.
- The meeting to consult shall be held within four weeks of the Dean's receipt of the written request from the faculty member. The faculty member may be accompanied by an academic colleague.
- 6.3.2.3 Following the meeting, the Dean and Faculty Committee shall each make a final recommendation on the case. If the final recommendations of both the Dean and Faculty Committee are negative, the Dean shall provide the faculty member with a written statement of reasons within one week of the meeting.
- 6.3.3 If the final recommendation of both the Dean and the Faculty Committee are negative, the faculty member may appeal as follows:
- 6.3.3.1 Faculty members in the Provost stream may appeal to the Faculty Committee on Appeals. (See section 7.2.)
- 6.3.3.2 Faculty members in the Senate stream may appeal to a University Grievance Committee. (See section 10.3.)
- 6.3.4 A negative decision of a Dean and/or Faculty Committee resulting from consideration of an initial appointment or reappointment (renewal of a Limited Term Appointment) may not be appealed.
- 6.3.5. A negative decision of a Dean and/or Faculty Committee resulting from consideration of a new appointment involving a change of stream may not be appealed; however, where there is an objection on procedural grounds, see section 10.1.2.

7 APPEALS COMMITTEE: COMPOSITION AND MANDATE:

7.1 CONFIDENTIALITY

- 7.1.1 All members of any committee mandated under these *Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments* shall respect the confidentiality of the committee's deliberations, consultations and any other relevant committee activities or proceedings.
- 7.1.2 This constraint shall also apply to all other individuals who may from time to time be required to appear before or otherwise be involved in the <u>in camera</u> proceedings of any such committee in the capacity of consultant, counsel or academic colleague, witness or party.

7.2 FACULTY COMMITTEE ON APPEALS:

- 7.2.1 There shall be a Faculty Committee on Appeals, elected by the Council of the Faculty, composed of:
 - (i) four Clinical Academics at the rank of Associate Professor or Professor with Continuing Appointments in the Provost stream, elected for a three-year term;
 - (ii) a slate of alternates to serve in cases where conflict of interest arises or because of unavailability.

NOTE: At least two of the members shall have been members of a Department Appointments and Promotions Committee or the Faculty Appointments and Promotions Committee within five years of election. None shall be a Department Chair. The Dean shall not be a member.

- 7.2.2 The Chair of the Committee shall be elected by the members to serve for a one-year term.
- 7.2.3 Under no circumstances shall a member of the Committee consider cases which they have already considered at another level or which arise from his or her home Department
- 7.2.4 No member of the Faculty Appointments and Promotions Committee shall at the same time be a member of the Faculty Committee on Appeals
- 7.2.5 Should any member of the Faculty Committee on Appeals have a conflict of interest, including the Chair, that member shall withdraw from consideration of the relevant case.
- 7.2.6 The Committee shall consider all appeals from faculty members in the Provost stream against negative decisions by the Dean and Faculty Committee regarding promotion and/or granting of a Continuing Appointment (other than negative decisions regarding an initial Continuing Appointment or a new Continuing Appointment involving a change of stream), or review of a Continuing Appointment.
- 7.2.7 Appeals to the Faculty Committee on Appeals shall be made in writing, stating all grounds of appeal, within four weeks from the date of the letter giving notification of the final decision giving rise to the appeal.
- 7.2.8 The Committee shall review all previously assembled documentation bearing on the case. In addition it may request statements from the faculty member, Department Chair, Department Committee, Dean or Faculty Committee.
- 7.2.9 The decision of the Committee is final.
- 7.2.10 The decision of the Committee shall be forwarded to the Dean and the appellant, with copies to the Provost and the Department Chair, accompanied by a statement of the reasons for the decision.

8 LEAVES:

8.1 Any faculty member is eligible to apply to the Department Chair or Dean, as appropriate, for a period of leave from academic responsibilities, including leave to pursue or renew research interests under the following conditions:

- 8.1.1 Arrangements for leave from hospital or institute responsibilities need to be negotiated with the hospital or institute;
- 8.1.2 Financial and resource arrangements need to be negotiated with all agencies involved in the appointment; and
- 8.1.3 All such leaves require approval from the Department Chair, Dean and Provost.

9 RESIGNATION, SUSPENSION AND DISMISSAL:

9.1 RESIGNATION:

- 9.1.1 The appointment and contractual arrangements applying to a faculty member may be terminated by the faculty member giving three months' notice in writing to the University.
- 9.1.2 Normally, all resignations shall become effective at the end of the academic year, June 30.

9.2 SUSPENSION:

- 9.2.1 Suspension is a condition in which a faculty member is temporarily debarred, in whole or in part, from his or her university position and/or the duties, functions and privileges pertaining to that position. Suspension is only possible pending a hearing.
- 9.2.2 A faculty member may be suspended by the President only when in his or her opinion the member's conduct poses a threat to the proper functioning of the University. In such a case, the President shall immediately inform the faculty member, in writing, of the reasons for and terms of the suspension.
- 9.2.2.1 Within seven days of a member's suspension, the President shall convene a Suspension Committee that shall terminate or confirm the suspension. Where the suspension is confirmed, the Suspension Committee shall make provisions for the resolution of the issue by referral to a hearing committee at an appropriate time as provided in section 9.3.5.1. Where the suspension is confirmed, the President shall complete the procedures described in 9.3.5 to 9.3.8.
- 9.2.2.2 Where the Suspension Committee confirms the suspension, it may recommend to the President that the suspended faculty member be permitted normal use of appropriate University facilities during the period of the suspension.
- 9.2.3 The Suspension Committee shall be composed of:
 - (i) the Vice-Chair of Senate who shall be Chair and who shall not vote:
 - (ii) the Dean of the Faculty of Medicine & Dentistry:
 - (iii) the Chair of the member's Department
 - (iv) three faculty members acceptable to the President and the person suspended. (If agreement cannot be reached, the three faculty members shall be elected by Senate.)
- 9.2.4 Should any member of the Committee, including the Chair, have a conflict of interest, he or she shall withdraw from consideration of the case.
- 9.2.5 Both the Provost and the faculty member suspended shall be invited to appear before the Suspension Committee.
- 9.2.6 The suspended faculty member shall have the right to appear with an academic colleague and/or with counsel.
- 9.2.7 Within five days of its final meeting on the case, the Suspension Committee shall terminate or confirm the suspension. The Suspension Committee shall report its decision to the President, in writing, with a copy to the faculty member.
- 9.2.8 The salary and other benefits of the faculty member shall continue throughout the period of suspension.

9.3 DISMISSAL:

- 9.3.1 Dismissal means the termination of an appointment by the University without the consent of the faculty member in the following circumstances:
- 9.3.1.1 In the case of a faculty member with a Limited Term Appointment, termination effective prior to the end of the specified period except for termination of an appointment in circumstances set out in section 3.1.6.
- 9.3.1.2 In the case of a faculty member with a Continuing Appointment, termination other than the termination of an appointment in any of the circumstances set out in sections 3.1.6 and 4.4.
- 9.3.2. None of the following constitutes dismissal:
 - (i) resignation;
 - (ii) the decision not to renew a Limited Term Appointment;
 - (iii) the decision not to grant a Continuing Appointment;
 - (iv) the decision to end a Continuing Appointment under section 4.4;
 - (v) termination of a Limited Term or a Continuing Appointment in circumstances set out in section 3.1.6;
 - (vi) retirement.
- 9.3.3 Dismissal may take place only upon proof of cause before a Hearing Committee. The precise meaning of cause in a particular case shall be determined by the Hearing Committee. "Cause" may include, but is not limited to, the following:
 - (i) gross misconduct;
 - (ii) persistent failure to carry out reasonable duties assigned by the appropriate academic authority;
 - (iii) failure to comply with the terms of the appointment as mutually agreed to in the letter of appointment or reappointment and the accompanying contract;
 - (iv) abandonment of duties.
- 9.3.4 Inability to carry out duties for medical reasons shall be treated separately from dismissal cases. Such separate treatment shall be identified as expeditiously as possible and with the assistance of external independent experts acceptable to the University.
- 9.3.5 In the case where:
- 9.3.5.1 either a suspension has been confirmed pursuant to 9.2.2.1 OR (in the cases where suspension does not precede dismissal proceedings)
- 9.3.5.2 the President, following consultation with the Provost and the Dean and the Chair of the Department concerned, is of the opinion that cause for dismissal exists; the President shall then initiate and complete the procedures below
- 9.3.6 The President shall notify the faculty member that grounds for dismissal have been alleged and (where applicable) that continuation of suspension and referral to a Hearing committee have been made, pursuant to 9.2.2.1. In either case, the president shall invite the faculty member to discuss the matter in the presence of the faculty member's Department Chair, the Dean, the Provost and, if the faculty member so wishes, an academic colleague.
- 9.3.7 If an attempt to settle the matter by this informal mediation fails, the President shall inform the faculty member in writing of the charges against him or her in sufficient detail to enable him or her to prepare his or her defence, and a reasonable period for such preparation shall be allowed.
- 9.3.8 As appropriate in each individual case, the President shall convene a Hearing Committee which shall determine whether cause for dismissal exists.
- 9.3.8.1 The Hearing Committee shall be composed of three members, either from inside or outside the University, acceptable to the President and the faculty member concerned.

9.3.9 The Hearing Committee procedures shall be: 9.3.9.1 The Hearing committee shall elect one of its own members as Chair. 9.3.9.2 The Chair of the Hearing Committee shall give the member of Faculty involved notice in writing of the time and place of the hearing. 9.3.9.3 The Chair of the Hearing Committee shall arrange that a verbatim record of the hearing be made. 9.3.9.4 The hearing shall be open to the public, except on the written request of the faculty member involved, or, where the Hearing Committee is of the opinion that (a) matters involving public security may be disclosed, or (b) intimate financial and personal matters may be disclosed at the hearing of such a nature, having regard to the circumstances, that the desirability of avoiding disclosure thereof in the interest of any person affected or in the University's or the public's interest out weight the desirability of adhering to the principle that hearings be open to the public. In cases involving such matters, the Committee may hold the hearing, or appropriate parts of it, in camera. All persons admitted to an in camera hearing shall respect the confidentiality of the proceedings. 9.3.9.5 Should any member of the Hearing Committee, including the Chair, have a conflict of interest, he or she shall withdraw from consideration of the case and be replaced in accordance with the procedures for the striking of the Hearing Committee as prescribed in 9.3.8.1. 9.3.9.6 The President and the faculty member shall be parties to the proceedings. 9.3.9.7 The President and the faculty member involved shall have the right to attend the hearing with counsel, provided that due notice is given to the Chair of the intention to present counsel. 9.3.9.8 The Hearing Committee shall hear such witnesses as may be called by the President and the faculty member concerned and shall entertain only evidence relevant to the charges as specified in 9.3.7. 9.3.9.9. Each party or his or her counsel may cross-examine any witness. 9.3.9.10 The Hearing Committee shall decide in camera whether or not the charges are substantiated. Where the Hearing Committee decides that the charges are substantiated, it shall decide whether cause for dismissal has been found, and, if so, it shall recommend dismissal. The decision of the Hearing Committee, accompanied by statement of reasons, shall be sent forward by the 9.3.9.11 Hearing Committee Chair to the President, with copies to the principal parties and the Provost, within two weeks of the Committee's final meeting on the case. 9.3.9.12 Where the Hearing Committee is not unanimous in its decision, the decision of the majority shall be the decision of the Committee. 9.3.9.13 The Chair of the Hearing Committee shall forward the decision to the President, who shall notify the faculty member of the decision. 9.3.9.14 The decision of the Hearing Committee shall be final. 9.3.10 Where the Hearing Committee recommends dismissal: 9.3.10.1 the President shall transmit the recommendation to the Board of Governors; 9.3.10.2 the University shall continue the salary and other benefits of the member of Faculty involved for three months

from the date of dismissal. The date of dismissal shall be the date on which the Board of Governors accepts

the recommendation for dismissal from the Hearing Committee, made under 9.3.9.10;

9.3.10.3 the University shall notify the appropriate hospital or institution.

- 9.3.11 Where the Hearing Committee finds cause for dismissal, the verbatim records of the hearing shall be transcribed and kept on file by the Secretary of Senate and shall be made available on request to the parties involved. These records shall be destroyed seven years after the date of dismissal.
- 9.3.12 Where the Hearing Committee does not find cause for dismissal, any suspension shall be lifted and the matter shall be regarded as closed. The President shall so inform the member of Faculty involved, and the records of the Hearing Committee shall be destroyed forthwith.

10 GRIEVANCE:

10.1 GENERAL:

- 10.1.1 Where a Clinical Academic considers that he or she has cause for grievance on any matter other than suspension, dismissal, promotion in Provost stream, a recommendation concerning granting of a Continuing Appointment in Provost stream, remuneration or salary (for which other procedures exist) e.g., assignment of teaching duties, amount of space or other facilities provided, support of research etc., he or she may grieve to a Faculty Grievance Committee. (See section 10.2.)
- 10.1.2 Where a Clinical Academic considers that he or she has cause for grievance on procedural grounds, i.e., other than on the merits, concerning determination of a request for change in stream, he/she may grieve to a Faculty Grievance Committee under section 10.2 or to a University Grievance Committee under section 10.3, as appropriate.
- 10.1.3 Where a Clinical Academic, Senate stream, considers that he or she has cause for grievance concerning promotion and/or the granting of a Continuing Appointment, he/she may grieve to a University Grievance Committee under section 10.3. In such a case, the time limits stipulated in 10.2.2 shall apply, 10.3.5 below notwithstanding.
- 10.1.4 Where a Clinical Academic considers that he or she has cause for grievance on any matter as described above but where that grievance involves a party from outside the member's home Faculty, he or she may grieve to a University Grievance Committee. (See section 10.3.) In such a case, the time limits stipulated in 10.2.2 shall apply, 10.3.5 below notwithstanding.
- 10.1.5 With the exception of grievances under 10.1.2 and 10.1.3 where final reasons have been provided, before making use of these grievance procedures, the aggrieved party shall first make reasonable efforts to resolve the difficulty through direct discussions with the other party and/or Dean.

10.2 FACULTY GRIEVANCE COMMITTEE

- 10.2.1 Grievances to a Faculty Grievance Committee* shall be submitted through the Dean of the Faculty, stating clearly the issue(s) under grievance giving detailed reasons for the grievance.
- 10.2.2 Such grievances shall be initiated within six weeks from the time the individual is notified, in writing, of the action or recommendations giving rise to the grievance.
- The Faculty Grievance Committee shall be composed of three Clinical Academics with Continuing Appointments, at the rank of Associate Professor or Professor, in accordance with the provisions of 10.2.2.2 and 10.2.2.3.
- The members of the Faculty Grievance Committee shall, on each occasion, be selected by lot from among a panel of at least nine Clinical Academics at the rank of Associate Professor or Professor, elected by Faculty Council.
- 10.2.2.3 Either of the principal parties to the grievance may request the replacement of any member of the Faculty Grievance Committee composed in accordance with 10.2.2.1 and 10.2.2.2 should either party consider that a conflict of interest exists. See also 10.2.5
- 10.2.3 The Committee shall elect its own Chair.

- 10.2.4 Before considering a grievance, the Committee shall determine whether the grievance falls within its jurisdiction. In a case where the Committee determines a particular grievance does <u>not</u> fall within its jurisdiction, it shall give a statement of reasons in writing to the parties to the grievance.
- 10.2.5 Should any member of the Faculty Grievance Committee including the Chair have a conflict of interest, that person shall withdraw from consideration of the relevant case or cases and his or her seat shall be declared vacant. In this event, in conformity with 10.2.2.2 and 10.2.2.3, an appropriate replacement shall be sought.
- 10.2.6 The Faculty Grievance Committee shall consider the issues and reasons cited in launching the grievance, the responses of those against whom the grievance has been launched, and other relevant information which the Committee obtains. The Committee shall make appropriate recommendations to resolve the grievance.
- 10.2.7 The Faculty Grievance Committee shall have reasonable access to the relevant information which the Committee considers necessary to the performance of the task.
- 10.2.8 Where the Committee so requests, members of Faculty are expected to render reasonable assistance to the Committee.
- 10.2.9 The Faculty Grievance Committee shall establish procedures appropriate to the circumstances of the case.
- 10.2.10 The Faculty Grievance Committee shall conduct its work as expeditiously as possible.
- 10.2.11 The parties to a grievance may be accompanied at the grievance by an academic colleague.
- 10.2.12 The Faculty Grievance Committee shall notify all parties of its decision in writing with reasons within one week of its final meeting on that case.
- 10.2.13 All those involved in a grievance shall respect the confidentiality of the information and proceedings.
- * The Faculty Grievance Committee may also hear grievances submitted to it by members of the academic staff on Limited-Duties Appointments but only in accordance with the relevant provisions of <u>Appointment Procedures: Members of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry.)</u>.

10.3 UNIVERSITY GRIEVANCE COMMITTEE:

- 10.3.1 If the Faculty Grievance Committee is not able to bring about a satisfactory settlement in a reasonable time, or if the recommendation of the Committee is not acceptable to the Department Chair, the Dean, or to the aggrieved party, any of the above parties may grieve in the second instance to a University Grievance Committee.**
- 10.3.2 Where a member of Faculty considers that he or she has cause for grievance on any matter as described in 10.1.1 but where that grievance involves a party from outside the member's home Faculty, he or she may grieve to a University Grievance Committee in the first instance, pursuant to 10.1.4 above. In such a case, the time limits stipulated in 10.2.2 shall apply, 10.3.5 notwithstanding.
- 10.3.3 Where a member of Faculty considers that he or she has cause for grievance on any matter as described in 10.1.3, he or she may grieve to a University Grievance Committee in the first instance. In such a case, the time limits stipulated in 10.2.2 shall apply, 10.3.5 notwithstanding.
- 10.3.4 Grievances to the University Grievance Committee shall be submitted through the, Director of Faculty Relations stating clearly the issues under grievance and giving detailed reasons for the grievance.
- 10.3.5 Except in the instance of a case brought forward under 10.1.3, 10.1.4 and 10.3.2, grievances to the University Grievance Committee shall be initiated within two weeks from the time when the parties receive in writing the decision of the Faculty Grievance Committee or the Faculty Grievance Committee written notice that the Faculty Grievance Committee has declared itself unable to bring about a satisfactory settlement.

- 10.3.6 The University Grievance Committee shall be composed of three members:
 - a) one tenured member not from the Faculty of Medicine & Dentistry chosen by the Provost;
 - b) one member who shall have a Continuing Appointment or hold a tenured appointment in the Faculty of Medicine & Dentistry or elsewhere within the University chosen by the grievor; and
 - c) one member who shall have a Continuing Appointment or hold a tenured appointment in the Faculty of Medicine & Dentistry or elsewhere within the University chosen by the Dean..
- 10.3.7 Either of the principal parties to the grievance may request the replacement of a member of the University Grievance Committee composed in accordance with the provisions of 10.3.6 should either party consider that a conflict of interest exists. See also 10.3.8.
- 10.3.8 Should any member of a University Grievance Committee have a conflict of interest, that person shall withdraw from consideration of the relevant case or cases and his or her seat on the Committee shall be declared vacant. In this event, in conformity with 10.3.6 and 10.3.7, an appropriate replacement shall be selected.
- 10.3.9 The Committee shall be convened initially by the Director of Faculty Relations.
- 10.3.10 The Committee shall elect its own Chair.
- 10.3.11 Before considering a grievance, the Committee shall determine whether the grievance falls within its jurisdiction.
- 10.3.12 In a case where the Committee determines a particular grievance does <u>not</u> fall within its jurisdiction, it shall give the President a statement of reasons in writing with copies to the parties to the grievance.
- 10.3.13 The University Grievance Committee shall consider the issues and reasons cited in launching the grievance, the responses of those against whom the grievance has been launched, and other relevant information which the Committee obtains. The Committee shall make appropriate recommendations to resolve the grievance.
- 10.3.14 The University Grievance Committee shall have reasonable access to the relevant information which the Committee considers necessary to the performance of the task.
- 10.3.15 Where the Committee so requests, members of Faculty are expected to render reasonable assistance to the Committee.
- 10.3.16 The University Grievance Committee shall establish procedures appropriate to the circumstances of the case.
- 10.3.17 The University Grievance Committee shall conduct its work as expeditiously as possible.
- 10.3.18 The parties to a grievance may be accompanied at the grievance by an academic colleague.
- 10.3.19 All those involved in a grievance shall respect the confidentiality of the information and proceedings.
- 10.3.20 The decision of the University Grievance Committee shall be final.
- 10.3.21 The University Grievance Committee shall report its decision in writing to the President with copies to all parties within two weeks of the termination of the hearing. The President shall institute appropriate measures and so inform the principal parties in writing.
- The University Grievance Committee may also hear grievances submitted to it by members of the academic staff on Limited-Duties Appointments but only in accordance with the relevant provisions of <u>Appointment Procedures: Members</u> of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry)..

THE UNIVERSITY OF WESTERN ONTARIO LONDON CANADA

APPOINTMENTS PROCEDURES: MEMBERS OF ACADEMIC STAFF WITH LIMITED DUTIES

(PHYSICIANS IN THE FACULTY OF MEDICINE & DENTISTRY)

APPROVED BY SENATE and BOARD OF GOVERNORS November 1984

(as amended 2002 for Faculty of Medicine & Dentistry)

THE UNIVERSITY OF WESTERN ONTARIO APPOINTMENTS PROCEDURES: MEMBERS OF ACADEMIC STAFF WITH LIMITED DUTIES (PHYSICIANS IN THE FACULTY OF MEDICINE & DENTISTRY)

A. Preamble:

All members of the academic staff of the University are appointed by the Board of Governors on the recommendation of the President. All policies and procedures followed in recommending appointments are established and recommended by Senate. The following policies and procedures shall specifically apply only to the appointment of academic staff with limited duties as defined in "Definitions" #5 below.

B. Definitions:

- 1. The "University" shall mean The University of Western Ontario as established in *The University of Western Ontario Act 1982* (as amended, 1988).
- 2. The terms "Senate", "Faculty", "Dean", "Department", "Chair", "academic staff", and related terms shall be interpreted in accordance with the provisions of The University of Western Ontario Act 1982 (as amended, 1988).
- 3. The term "Appointments Committee" refers to the Department Appointments and Promotions Committee whose structure and composition are set out in Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments (1999) as amended 2002. In the case of Limited-Duties Appointments, the role of this Committee shall be limited to the relevant provisions in Section D, "Appointments Process" of Appointments Procedures: Members of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry).
- 4. The terms "Faculty Grievance Committee" and "University Grievance Committee" refer to the Committees of that name whose structure and composition are set out in Conditions of Appointment: Physicians Appointed in Clinical Departments and Clinical Divisions of Basic Science Departments (1999) as amended 2002, Section 10 "Grievance." In the case of Limited-Duties Appointments, the jurisdiction of these committees shall be limited relevant to the provisions of Section E. "Grievance", of Appointments Procedures: Members of Academic Staff with Limited Duties (Physicians in the Faculty of Medicine & Dentistry).
- 5. A "Limited-Duties Appointment" is an appointment of a physician to the academic staff of the University where the appointee carries less than a full load of teaching and other duties as determined by the Dean of the Faculty and such duties are specified in the Letter of Appointment.
- 6. An "Adjunct Professor" is a physician who:
 - (a) is generally community-based, but may be located in an affiliated hospital or institution;
 - (b) does not otherwise hold an academic appointment at the University (with the exception of other Adjunct or Limited-Duties Appointments);
 - (c) performs specific limited responsibilities relating to teaching and/or research ancillary to the appointee's principal occupation;
 - (d) generally receives no remuneration, although there may be a small stipend, and
 - (e) is not required to participate in a departmental practice plan.
- 7. An "Honorary Professor" is a physician who:
 - (a) has attained academic or professional distinction in his or her field;
 - (b) does not otherwise hold an academic appointment at the University (with the exception of other Honorary Appointments)
 - (c) is not required to perform any teaching or other academic responsibilities;
 - (d) receives no remuneration, and
 - (e) is not required to participate in a departmental practice plan.

Appointment as an Honorary Professor recognizes a mutually beneficial scholarly association between the University and the appointee.

8. A "Letter of Appointment or Reappointment" is the letter to the prospective appointee signed by the Dean of the Faculty. This document specifies the academic contract type (Limited-Duties), the status of the contract (Regular or Post-Retirement), the starting and end date of the appointment, the academic rank (Adjunct Professor or Honorary Professor) and the expected duties of the appointee.

C. Terms and Conditions:

1. A Limited-Duties Appointment shall be for a fixed term. The status of the appointment shall be:

```
Regular (i.e., up to five years)
Post-Retirement (i.e., up to three years)
```

2. The following academic ranks shall be used:

Adjunct Professor Honorary Professor.

- 3. The initial appointment or repeat appointment as the case may be shall take into account the appointee's
 - (i) qualifications
 - (ii) experience, and
 - (iii) achievements

and shall be included as part of the recommendation or recommendations in paragraph 1 below in Section D, "Appointments Process".

- 4. A Limited-Duties Appointment does not carry tenure or Continuing Appointment status nor shall an appointee be considered eligible for tenure or a Continuing Appointment.
- 5. A Limited-Duties Appointment, in the first instance, cannot extend past the fifth anniversary of the effective date of the appointment. Thereafter, a Limited-Duties Appointment may be repeated
- 6. A Limited-Duties Appointment cannot extend past the effective date of retirement. A Limited-Duties Appointment following retirement will be a post-retirement Limited-Duties Appointment.
- 7. Further to paragraph 5 above, a Limited-Duties Appointment may be repeated provided that, in each instance, the extension shall not extend past the fifth anniversary of the effective date of the repeat appointment.
- 8. Notwithstanding that a Limited-Duties Appointment may be repeated, further appointments beyond the initial appointment are neither promised nor guaranteed.
- 9. On the recommendation of the President, the Board of Governors may terminate a Limited-Duties Appointment during the period of the appointment should the standard of conduct or performance of duties fall below the minimum acceptable standard for academic staff of the University.

D. Appointments Process:

- 1. The appointments process shall be as follows:
 - Subject to authorization by the Dean to make an appointment, the Department Appointments and Promotions Committee may recommend proposed Limited-Duties appointments (or repeat appointments).
 - ii) The recommendation of the Appointments Committee shall be sent forward to the Dean by the Department Chair. The Chair shall, at the same time, send forward his/her own recommendation.
 - iii) The Dean shall send forward to the Provost and Vice-President (Academic), the recommendation of the Department Appointments and Promotions Committee together with that of the Department Chair. The Dean shall, at the same time, send forward his/her own recommendation.

- 2. It is recognized that the Department Chair may find it necessary to forward a recommendation for a proposed Limited-Duties Appointment unaccompanied by a recommendation from the Appointments Committee. In such cases, the Chair shall report such an appointment to the Appointments Committee. Recommendations made under this paragraph shall not conflict with any of the provisions of this document under "Definitions", "Preamble", and "Terms and Conditions" above.
- The Appointments Committee may delegate the task of making Limited-Duties Appointments to the Department Chair. Such delegation will be reviewed on an annual basis.

E. Grievance:

(Faculty)

- 1. There shall be no right of hearing at any level of the appointment process. However, where a member of the academic staff on a Limited-Duties Appointment considers that he/she has cause for grievance in respect to the terms and conditions in a letter of appointment written on behalf of the University not being met, he/she may, after first making reasonable efforts to resolve the difficulty through direct discussions with his/her Dean, grieve to the Faculty Grievance Committee.
- 2. Grievances to the Faculty Grievance Committee shall be submitted through the Dean of the Faculty stating clearly the issue(s) under grievance and giving detailed reasons for the grievance.
- 3. Such grievances shall be initiated within three weeks from the time when the individual is notified, in writing, of the action or recommendation giving rise to the grievance. This time-limit is mandatory.
- 4. The Faculty Grievance Committee shall notify all parties of its recommendation in writing with reasons within one week of its final meeting on that case.

(University)

- 5. If the Faculty Grievance Committee is not able to bring about a satisfactory settlement in a reasonable time, or if the recommendation is not satisfactory to the Department Chair, to the Dean, or the academic staff member, either party may grieve, in the second instance, to the University Grievance Committee.
- 6. Grievances to the University Grievance Committee shall be submitted through the Director of Faculty Relations stating clearly the issue(s) under grievance and giving detailed reasons for the grievance.
- 7. Grievances shall be initiated within two weeks from the time when the parties receive in writing the decision of the Faculty Grievance Committee or written notice that the Faculty Grievance Committee has declared itself unable to bring about a satisfactory settlement. This time limit is mandatory.
- 8. The University Grievance Committee shall notify all parties of its recommendation in writing with reasons within one week of its final hearing on that case.
- The decision of the University Grievance Committee shall be final insofar as Senate procedures are concerned.

Board of Governors APPENDIX VIII May 9, 2002 Annex 4

Report of the 259th Meeting of the Council of Ontario Universities Thursday and Friday, February 14 and 15, 2002

D.M.R. Bentley, Academic Colleague

At the meetings of the Academic Colleagues and Council discussions focused on four matters that will be of interest to Senate.

(1) Enrolment Predictions and Funding Commitments by the Provincial Government. On the basis of recent information, the increase in first-year enrolment in 2002-03 has been revised from 1.6% to 3.4% (approximately 4,500 more students) and in 2003-04 from 3.7% to 4.9% (approximately 6,800 more students). Because the Government's financial allocations were based on the earlier estimate, these figures indicate that, in order fully to fund the growth in the system, existing budget commitments need to be increased by approximately \$30,715,000 to cover 2002-03 and \$46,185,000 to cover 2003-04. COU is attempting to address this situation in two ways: (1) by requesting further funds from SuperBuild; and (2) by requesting additions to base budget to fund enrolment growth and to offset inflation. Given the unlikelihood that the Government will provide either all or nothing of what is being requested (financial constraints militate against the former, public pressure in response to the "double cohort" against the latter), Ontario's universities will almost certainly find themselves in their usual position of having to make do with an amount that makes both loud complaint and unmitigated jubilation impossible.

The matter is complicated by the PC leadership race, which will not be decided until March 23. Very clearly, the two candidates who are emerging as front-runners, Ernie Eves and Jim Flaherty, are Conservatives of quite different stripes and agendas, not least with respect to post-secondary education. (There may be some comfort to be drawn form the fact that Michael Gourley, once again displaying the capacity for self-transformation that he perhaps acquired during his brief stint as Polkaroo, is Chief Policy Advisor to Mr. Eves.) To assist in clarifying the educational agendas of all the leadership candidates, COU has prepared a list of questions, their answers to which will be made publicly available.*

(2) Capacity and Choice. In view of the greater than predicted increase in enrolment that seems certain to occur in 2002-03 and 2003-04, it is by no means certain that the Ontario university system can accommodate all the students who have applied for entrance or that, even if it could, as many students as usual (42%) would be able to get their first choice of university and programme. By the time Senate receives this Report, wide publicity may well have been given to the University of Toronto's statement to secondary-school counselors to the effect that it will have difficulty in meeting the higher demands outlined in (1), above without further increases in government funding. (The final two sentences of the Toronto statement are worth quoting to convey a sense of its careful tone: "We acknowledge that government, through the Budget and through the SuperBuild Growth Fund, has made a significant

start toward providing the resources needed [to cope with the 'double cohort']. As a shared investment in the students and families of Ontario, we look forward to receiving a commitment from government for the operating and capital support that will enable us to offer spaces to the larger number of students seeking a place at the University of Toronto.")

- (3) Marks Comparability. On the basis of the limited data available from Grade 10, there appears to be less disparity between the marks being awarded to students in the "old" and new curricula than some reports in the media have suggested. Across the system as a whole, students taking the new curriculum are getting marks that are very slightly lower in some subjects (such as Mathematics) and slightly higher in others (such as French and History); however, there may well be wide divergences in the practices of individual teachers, schools, and school boards, and there is some evidence to suggest that students in the Applied stream are having greater difficulty with the new curriculum than students taking University and University/College courses.
- (4) **University of Ontario Institute of Technology.** COU will tactfully attempt to persuade the Government of the folly of using the phrase "University of Ontario" in the bill (139) that will transform this particular egg into a caterpillar. It will also be attempting to ensure that, whatever it is called, the emerging institution will have a senate (or equivalent) as part of its governance structure and that its academic staff will have the protections normally accorded to university faculty.

All in all, the times seem to require from the universities and from Council a combination of nimble footwork, handkerchief fluttering, and stick brandishing that is distressingly similar to Morris dancing.

*

Since this report was written, the COU letter has been dispatched to the candidates and Mr. Eves has replied that he is "proposing a second phase of the Ontario Student Opportunities Fund, to build on the commitment [he] made in a previous Budget" and stating his belief "that special provisions must be made and will be made to address the double-cohort issue."

Board of Governors

May 9, 2002

APPENDIX VIII

Annex 5

REPORT OF THE HONORARY DEGREES COMMITTEE

FOR INFORMATION

Honorary Degree Recipients - Spring Convocation 2002

The Honorary Degrees Committee of the Senate announces the following persons who are to be honored by conferment of degrees *honoris causa* at the 278th Convocation of the University to be held Tuesday June 4 to Friday, June 7, 2002.

Tuesday, June 4 Faculty of Graduate Studies*

10:00 a.m. Faculty of Arts

Linda Hutcheon - D.Litt. Faculty of Information and Media Studies

Faculty of Music

* = students in programs hosted by the Faculties of Arts, Information

and Media Studies and Music

Tuesday, June 4 Faculty of Graduate Studies*
3:30 p.m. Richard Ivey School of Business

Henry King, Jr. - D.C.L. Faculty of Engineering

Faculty of Law

Faculty of Medicine & Dentistry

* = students in programs hosted by the Richard Ivey School of Business,

and the Faculties of Engineering, and Medicine & Dentistry

Wednesday, June 5 Faculty of Graduate Studies*

10:00 a.m. Faculty of Education

Vandana Shiva, LL.D. * = students in programs hosted by the Faculty of Education

Wednesday, June 5 Brescia University College 3:30 p.m. Brescia University College

Shelagh Rogers - LL.D. King's College

Thursday, June 6 Faculty of Graduate Studies*

10:00 a.m. Faculty of Social Science (Honors and ACS)

Mel Cappe - LL.D. * = students in programs hosted by the Faculty of Social Science

Thursday, June 6 Faculty of Graduate Studies*

3:30 p.m. Faculty of Science

Kenneth Davey - D.Sc. * = students in programs hosted by the Faculty of Science

Friday, June 7 Faculty of Social Science (3 year programs)

10:00 a.m.

Pierre Berton - LL.D.

Friday, June 7 Faculty of Graduate Studies* 3:30 p.m. Faculty of Health Sciences

Lillian Fuller - D.Sc. * = students in programs hosted by the Faculty of Health Sciences